Author's response to reviews

Title: Quality of neonatal healthcare in Kilimanjaro region, northeast Tanzania: Learning from mothers’ experiences and challenges in neonatal care.

Authors:

Bernard Mbwele (b.mbwele@kcri.ac.tz)
Nicole Ide (niclea@gmail.com)
Elizabeth Reddy (elizabeth.reddy@duke.edu)
Sarah A. P. Ward (ward525@gmail.com)
Joshua A. Melnick (melnick.josh@gmail.com)
Rachael Manongi (rmanongi@yahoo.co.uk)

Version: 2 Date: 6 December 2012

Author's response to reviews:

Dear editors,

In this revision we have cited one of our recent Publication.

Few changes have been made in the area of Authors contribution and authors information.

I remain,

B. MBWELE